DR. PAUL L SWIECICKI (Orcid ID : 0000-0003-2286-3555)



Hematology/Oncology, Ann Arbor, Michigan ²Ann Arbor Veterans Affairs Medical Center, Department of Internal Medicine, Division of Hematology/Oncology, Ann Arbor, Michigan ³Rogel Cancer Center, University of Michigan Medical School, Ann Arbor, MI ⁴University of Michigan School of Public Health, Department of Biostatistics, Ann Arbor, Michigan ⁵University of Michigan Medical School, Department of Otolaryngology-Head and Neck Surgery, University of Michigan Health System, Ann Arbor, Michigan

This is the author manuscript accepted for publication and has undergone full peer review but has not been through the copyediting, typesetting, pagination and proofreading process, which may lead to differences between this version and the <u>Version of Record</u>. Please cite this article as <u>doi:</u> 10.1002/CNCR.33226

Axitinib in head and neck cancer

Corresponding Author

Paul L. Swiecicki, MD pswiecic@med.umich.edu C340, 1500 E. Medical Center Dr Ann Arbor MI 48109-5848 Telephone: 734-647-1017 Fax: 734-647-8689

Funding Statement

This study was approved and funded in part by the National Comprehensive Cancer Network (NCCN) Oncology Research Program from general research support provided by Pfizer, Inc

Conflict of Interest Statement

P. L. S. and J. C. B. have applied for provisional patents* (not awarded yet) for companion diagnostics related to the data presented in this publication. The remainder of the authors declare no potential conflicts of interest.

Author Contributions Statement

Paul Swiecicki: Conceptualization, data curation, formal analysis, methodology, resources, manuscript writing, and editing. **Emily Bellile**: Data curation, formal analysis, methodology, manuscript writing, and editing. **Collin Brummel**: Data curation, formal analysis, and manuscript editing. **J Chad Brenner**: Formal analysis, methodology, manuscript writing, and editing. **Francis Worden**: Conceptualization, resources, and manuscript editing.

Acknowledgements

We are grateful to the patients and families without whom this study would not have been possible. We are particularly appreciative to the clinical and research staff assisting with the care of these patients.

Lay Summary

Metastatic head and neck squamous cancer is an incurable disease which has limited treatment options and a poor prognosis. In this study, we are the first to demonstrate the targeted oral drug axitinib improves survival in patients with heavily pre-treated metastatic head and neck cancer. Furthermore, we observed that patients whose tumors had specific mutations derived the greatest benefit from from therapy. Investigation of axitinib in a genomic biomarker selected population alone or in combination with immunotherapy is warranted.

Precis for Table of Contents

This manuscript demonstrates that treatment with axitinib improves survival in patients with heavily pre-treated R/M HNSCC, alternate response criteria enables identification of patients with atypical radiographic responses, and patients with PI3K pathway alterations may derive exceptional benefit from therapy. Clinically, this study provides evidence for evaluation of axitinib in a genomic biomarker selected population alone or in combination with immunotherapy.

Abstract

Background: There are limited treatment options in unresectable recurrent or metastatic head and neck squamous cell carcinoma (R/M HNSCC). Vascular endothelial growth factor is of significant interest for targeted therapy in R/M HNSCC given its central role in tumorigenesis and immune suppression. Axitinib is a potent inhibitor of VEGFR1-3, PDGFR, and c-kit and offers such an approach. <u>Methods</u>: We report the results of a phase II trial evaluating axitinib in R/M HNSCC using the Choi Criteria for radiographic response assessment. The primary endpoint of this trial was 6 month overall survival.

<u>Results</u>: Twenty-nine patients were enrolled and 28 were evaluable for response. Patients were heavily pre-treated with 61% having had at least one previous systemic treatment in the metastatic setting (range 0-5). The median overall survival of 9.8 months with a 6 month overall survival was 70% which met the protocol defined criteria for clinical efficacy. Best overall response rate was 42%. Correlative analyses demonstrated that PI3K signaling pathway alterations were associated with an increased response to therapy (75% versus 17%). A marked response to therapy was seen in a subgroup of patients who were treated with an immune checkpoint inhibitor after progression on axitinib. <u>Conclusions</u>: Treatment with axitinib is associated with improved survival in patients with heavily pre-treated head and neck cancer and PI3K pathway alterations may serve as a biomarker for response. Further investigation is warranted to evaluate axitinib in biomarker selected populations, especially in combination with immune checkpoint inhibitor therapy.

Keywords:

Head and neck cancer, axitinib, vascular endothelial growth factor receptor inhibitor, PI3K, Choi Criteria

Total number of

1) Text pages: (including title page(s), abstract, main text, references, and figure legends): 19

- 2) Tables: 3
- 3) Figures: 3
- 4) Supporting files for publication: 1 Supplemental Table

<u>Main Text</u>

Introduction

Head and Neck Squamous Cell Carcinoma (HNSCC) is the 6th most common cancer with 600,000 new cases worldwide each year with an incidence rate that is increasing at an unprecedented rate due to the high prevalence of human papilloma virus (HPV)-induced HNSCC¹. In fact, oropharyngeal cancer is one of only four cancers increasing in incidence in the United States². Although the majority of patients with HNSCC are cured with multimodality therapy, a significant proportion develop unresectable recurrent or metastatic HNSCC (R/M HNSCC). Despite the recent development of programmed death-1(PD-1) inhibitors, response rates remain low due to variability within the immune micro-environment³. Even with these novel therapies, the median survival for patients newly diagnosed with R/M HNSCC is approximately 12 months⁴.

With increasing molecular characterization of head and neck squamous cell carcinoma, there has been significant interest in targeted therapy⁵. Vascular endothelial growth factor (VEGF) dysregulation has been identified as a crucial process in R/M HNSCC in not only angiogenesis, but also progression, immunosuppression, and immune tolerance^{6, 7}. Furthermore, VEGF overexpression is associated with advanced disease and poor prognosis^{8, 9}. Given this central role in advanced disease and tumorigenesis, VEGF inhibition is of significant interest as a candidate for targeted therapy.

Axitinib is a multi-receptor tyrosine kinase approved in renal cell carcinoma which inhibits several isoforms of VEGF receptor (VEGFR 1, 2, and 3). Furthermore, it has inhibitory activity against PDGFR and downstream effectors of EGFR both of which are commonly disrupted and contribute to head and neck tumorigenesis^{5, 10, 11}. Given this mechanism of action and known molecular alterations in R/M HNSCC, it seems to be a promising agent for clinical assessment.

We previously reported a phase II study evaluating axitinib in patients with heavily pretreated R/M HNSCC. This work demonstrated a low response (7%) rate with single agent axitinib, however a significant proportion of patients had stable disease (70%) with radiographic findings consistent with treatment response¹². Moreover, the population had an impressive overall survival (10.9 months) suggesting that efficacy was perhaps was not captured. Hence we postulated that axitinib held significant anti-tumor activity in R/M HNSCC but RECIST criteria failed to appropriately capture responders and may inappropriately suggested tumor progression. Differential manifestations of response have been seen with the use of tyrosine kinase inhibitors (ie swelling, cystic attenuation) which have the potential of abhorrently being interpreted as progressive disease by RECIST prompting the development of the Choi Criteria¹³.

Based on these findings, we initiated a new follow up phase II study to investigate the clinical activity of axitinib in R/M HNSCC using the Choi Criteria for response assessment. Our hypothesis was that axitinib would have significant anti-tumor activity as judged by the Choi Criteria and result in an improvement in the 6 month overall survival compared to a historical control.

Materials and Methods

Patient eligibility

This was a phase 2 open label trial approved by the Institutional Review Board (IRBMED) of the University of Michigan Rogel Cancer Center (NCT02762513). All patients provided written informed consent. Patients \geq 18 years old with histologically documented unresectable recurrent or metastatic head and neck squamous cell carcinoma were eligible. All patients were required to have the presence of measurable disease by CT scan or cutaneous lesions \geq 10 mm not assessable on imaging but present on physical exam, ECOG performance status of 0-2, and life expectancy of \geq 12 weeks. Adequate

hematopoietic, hepatic, and renal function were required and defined as: absolute neutrophil count \geq 1.5x10⁹ cell/ml, platelets \geq 75,000 cells/mm³, hemoglobin \geq 9.0 g/dL, concentrations of total serum bilirubin within 1.5x the upper limit of normal (ULN), aspartate aminotransferase (AST) and alanine aminotransferase (ALT) within 2.5x institutional upper limits of normal unless there were liver metastases in which case AST and ALT within 5.0 x ULN, serum creatinine clearance \geq 30 ml/min, urinary protein < 2+). Women of childbearing potential must have had a negative serum or urine pregnancy test within 3 days prior to treatment.

Patients with tumors encasing major blood vessels, active hemoptysis (> ½ teaspoon of bright red blood per day), or currently using therapeutic anticoagulation were excluded as were those with gastrointestinal abnormalities resulting in impaired absorption. Treatment with epidermal growth factor receptor inhibitors within 30 days preceding study entrance was prohibited. Patients were excluded if they had uncontrolled hypertension prior to enrollment which was defined as a systolic blood pressure readings >140 mm Hg and/or a diastolic blood pressure readings > 90 mm Hg.

Treatment plan

Enrolled patients underwent a complete history and physical examination, baseline laboratory studies (CBC with differential, comprehensive metabolic profile, TSH, urinalysis) and radiographic staging studies (CT Neck/Chest and others as clinically warranted). If cutaneous lesions were not assessable for response by imaging, pictures of the target lesion(s) were obtained as well. All screening assessments were completed within 28 days prior to the start of treatment.

Patients were initiated on axitinib 5 mg twice daily with a cycle length of 28 days. Dose escalation was planned at 2 weeks (to 7 mg twice daily) and 3 weeks to goal of 10 mg twice daily in the absence of grade 2 or greater toxicities. Patients were seen for toxicity assessment and laboratory assessments (CBC, CMP, TSH, UA) at two weeks, four weeks, and then monthly after treatment initiation. Response to therapy. Dose escalation could be resumed at the next visit if toxicities diminished to grade 1 or less. Treatment was continued until disease progression, unacceptable toxicity, patient withdrawal of consent, or investigator discretion.

Evaluation of response

Response assessment was performed after two cycles of axitinib treatment and continued every two cycles. Radiographic assessments obtained at enrollment were obtained at each time point. Similarly, if physical exam was being used for response assessment of cutaneous lesions, pictures were taken at each time point. Photographs as well as imaging studies were submitted to the University of Michigan Tumor Response and Assessment Core. Radiologic response was determined according to the Choi Criteria¹³.

Statistical considerations

Twenty nine patients were enrolled between 8/30/2016 and 10/23/19. The median follow up duration among the study participants was 18 months (range: 1-36) and no patients remain on therapy. Follow-up on patients still alive ranges from 5 to 32 months. Based on our previous study supporting an improvement in survival in patients with R/M HNSCC treated with Axitinib, we designed this expansion study. Although consideration was given to adjusting this original study to a Bayesian expansion trial design, it was ultimately decided to begin a new cohort to test for an improvement in survival under the same assumptions of mortality rate as the previous study. Of note, treatment continuation decisions for this trial were based on Choi criteria that, as previously reported¹², considerably differed from RECIST decisions when evaluated in the original trial.

The primary aim was to compare 6-month overall survival after treatment with Axitinib in patients with unresectable, recurrent or metastatic head and neck cancer to historical rates. Based on results in the literature, we assumed a 6-month mortality rate of 50% under current standard care in this patient population¹⁴. A sample size of 37 patients was planned to test whether survival after treatment with Axitinib is improved to 70% at 6-months compared to 50% with an upper tailed test of binomial proportion. No interim analyses for activity were planned. Based on observed clinical benefit and slowed accrual rate, an unplanned interim analysis was performed after enrollment of 29 patients. Data was analyzed by the study statistician; a statistically meaningful improvement in survival was identified in this analysis and the decision was made to close to further accrual.

Overall survival (OS) was defined as the time from study enrollment to death from any cause. Six month overall survival was the proportion of patients who received at least one cycle of Axitinib alive 6 months

after study enrollment and 95% confidence intervals were estimated using the Wilson score interval method. Treatment-related adverse events were graded according to the Common Terminology for Adverse Events version 4.03. Response rate was defined as the sum of patients with complete response (CR) and partial response (PR) per the Choi Criteria. Statistical analysis was performed with SAS v14.3 software (Carey, NC). Planned correlative analyses included genomic analysis of patients where next generation sequencing results were evaluable.

Results Patient characteristics

Twenty nine patients were enrolled, one of which died prior to treatment with axitinib. All twenty eight patients who received at least one dose of axitinib were included for toxicity analysis of which the baseline characteristics are summarized in Table 1. The mean age was 63.9 years old (range: 37-80) and the majority of patients (61%, n=17) had an ECOG performance score of 1 indicating mild impairment. The primary site of disease for most patients was the oropharynx (42.6%, n= 13) and the majority of study participants were HPV negative (60.7%, n=17). The majority of patients (61%, n=17) had at least one previous systemic treatment in the metastatic setting with the number or previous lines of treatments ranging from 0-5. Seventeen patients (61%) were refractory to platinum therapy defined as progression within 180 days of chemotherapy and twelve patients (43%) were previously treated with a PD-1 inhibitor.

<u>Toxicity</u>

The median duration of treatment was 3 cycles (range: 1-9).The most common toxicities included fatigue (75%), hypertension (54%), nausea (32%) and diarrhea (25%) (Table 2). Bleeding was observed in 5 patients, including one patient with a grade 3 lower GI bleed, all of which spontaneously resolved and did recur with re-initiation of axitinib. Grade 3 or 4 severe toxicities were seen in 16 patients (57%). Severe toxicities included fatigue (21%), hypertension (7%), and mucositis (7%). No grade 5 events were reported. Overall, observe toxicities were consistent with that previously reported in the literature^{15, 16}.

Efficacy

The 6 month overall survival was 71% (95% CI: 53-85%) (Table 3). This met the protocol defined criteria for supporting evidence of clinical benefit. The median progression free survival was 3.5 months (95% CI: 2.4-5.4 months) and median overall survival was 9.8 months (95% CI: 5.9-12.2 months) (Figure 1).

Three patients completed trial participation prior to response imaging; one due to adverse effects but was clinically noted to have progressive disease, one of whom died due to progressive disease, and a third whom withdrew from the study. The overall response rate was 42% and a disease control rate of 53%. The waterfall plot in Figure 2A graphically demonstrates the depth of response amongst participants evaluable for response. One patient had a durable complete response. Only one patient with cutaneous squamous cell carcinoma demonstrated a response to therapy. This patient had a mutation in KDR (VEGFR2) and achieved a durable complete response. All of the remaining six patients with cutaneous squamous cell carcinoma had progressive disease.

Given the immunomodulatory potential of VEGFR inhibition we evaluated the treatment response in patients who received PD-1 inhibition as part of their treatment course. Eleven patients were treated with a PD-1 inhibitor prior to treatment with axitinib. Three patients had primary resistance to checkpoint inhibitor therapy, none of whom responded to axitinib (0/3). Eight had acquired resistance to checkpoint inhibitor therapy of which 3 patients had a partial response with axitinib (3/8, 37%), 2 had stable disease (2/8, 25%), and 3 had progressive disease (3/8, 37%). Eleven patients were treated with a PD-1 inhibitor post progression on axitinib with an observed response rate of 45% (5/11). Response assessment demonstrated complete response in one patient (1/11, 9%), partial response in four patients (4/11, 36%), stable disease in one patient (1/11, 9%), and progressive disease was seen in the remaining five (5/11, 45%).

Correlative studies

To evaluate the association between genomic alterations, tumor characteristics, and clinical outcomes we analyzed results from patients who had commercial next generation sequencing previously performed (n=20). The investigators defined a set of genes (sequenced as part of all NGS panels) and recurrent alterations are shown (Figure 3). Importantly, while no mutations were identified in FLT1 (VEGFR1), FLT4 (VEGFR3), PDGFR or KIT, two patients had mutations in KDR (VEGFR2) including a S1100F

mutation as well as a patient with two mutant alleles R1032Q and G638R (Supplemental Table 1). The ability of axitinib to inhibit these mutant forms of KDR is unknown; however, the patient with the S110F mutation had a complete response whilst the other had progressive disease. Importantly, 55% (11/20) of the patients had TP53 alterations, 40% (8/20) of the patients harbored alterations to genes in the PI3K pathway, including PTEN and PIK3CA, and, 30% (6/20) of the patients had mutations in either KMT2C (MLL2) or KMT2D (MLL3).

The degree of response and pathway alterations were correlated for exploratory analysis (Figure 2B). The relative response rate for patients with mutations in the PI3K pathway was 75% vs 39% in those which were wild-type (6/8 versus 2/12 patients). In terms of the KMT2C/D pathway, the response rate was 33% in the mutant population versus 50% in the rest of the population (2/6 versus 6/12 patients). Given the differential responses seen between patients with cutaneous squamous cell carcinoma versus non-cutaneous primaries, the response rates were further explored (Table 3). Although sample sizes were limited, mutations in the PI3K pathway were associated with a higher response rate than the wild type population in the non-cutaneous squamous cell carcinoma (86% versus 12%).

Discussion

In this phase 2 study of patients with heavily pretreated unresectable recurrent or metastatic head and neck squamous cell carcinoma (R/M HNSCC), axitinib demonstrated an improvement in 6 month overall survival compared to a historical controls (70% vs 50%). Furthermore, treatment resulted in significant response rates and lower rates of severe toxicities.

There is increasing recognition of variable radiographic manifestations of response with the advent of novel classes of therapeutics. Most recognized is the 'pseudoprogression' observed with immunotherapy which prompted development of iRECIST to capture atypical responses¹⁷. The Choi Response Criteria have been best evaluated in gastrointestinal stromal tumors (GISTs) where, compared to RECIST, they have been demonstrated to better predict survival¹³. In our previous trial using RECIST we observed a low RECIST assessed objective response rate, but a paradoxically high impressive overall survival in heavily pre-treated patients¹². As such, we hypothesized that we were underappreciating treatment responses with the use of RECIST and the Choi Criteria may be more appropriate for discerning patients deriving benefiting from therapy. With the utilization of the Choi Criteria in this study, we identified a response rate of 42% with additional 11% having stable disease. Furthermore, use

of these response criteria for treatment decisions resulted in an improvement in overall survival compared to historical controls supporting both that the Choi Criteria appropriately identified treatment responders and that axitinib is an effective therapeutics in heavily pretreated R/M HNSCC.

Targeted therapy has the demonstrated promise in pre-clinical studies in HNSCC. Alterations in PI3KCA, CDKN2A, and EGFR suggest head and neck cancer being a candidate for development of targeted therapeutics. However, this approach has had limited clinical success. The only approved agent, cetuximab, has been demonstrated to improve survival by less than three months^{18, 19}. Tyrosine kinase inhibitors offer the benefit of targeting numerous pathways (ie VEGFR, EGFR, PDGFR) and isoforms simultaneously. Axitinib has been demonstrated to inhibit VEGFR (-1,-2, and -3) as well as c-Kit. Mounting evidence suggest that VEGF inhibition is immunomodulatory via numerous mechanisms including production of IFNy, reversal of the immunosuppressive microenvironment, and augmented activity of CD8+ T cells via hypoxia-inducible factor- 1α secondary to tumor hypoxia²⁰⁻²². As VEGFR inhibition may prime the immune system for response to immunotherapy, sequential use may be a modality to decrease toxicities yet still gain therapeutic synergy. In the small subgroup of patients that were treated with immunotherapy following axitinib (n=11), the RR to PD-1 monotherapy was 45% including one patient with a complete response. Although conclusions cannot be drawn given the limited sample size, previous trials have shown a RR of 13-17% in biomarker unselected populations^{4, 23,} ²⁴ hence supporting possible potentiation with sequential therapy. Preliminary results from the phase Ib/II KEYNOTE-526 trial evaluating concurrent lenvatinib (an inhibitor of VEGFR 1-3, FGFR 1-4, and PDGFRα) and pembrolizumab demonstrated a response rate of 40.9% and median PFS of 8.2 months supporting further investigation of this combination²⁵.

57% of patients in this study experienced grade 3 or 4 toxicities of which the most common was fatigue. Large studies of single agent treatment regimens employed in this patient population have demonstrated toxicity rates ranging from 35-46%^{18, 24}. Although this study has a higher rate of serious toxicities relative to comparable agents, our previous study of single agent axitinib demonstrated a much lower rate of severe toxicities (40%)¹². The toxicities encountered were manageable with dose reductions supporting patient tolerability. As aforementioned, there is promise of significant synergy with the combination of VEGF inhibition and PD-1 inhibition. Ongoing phase 3 trials are evaluating concurrent lenvatinib and pembrolizumab in patients with PD-L1 \geq 1%. However, preliminary reports of clinical trials evaluating this combination describe grade 3 or 4 toxicities in 91% of patients and leading to 18% of study participants discontinuing treatment. Sequential therapy (ie axitinib followed by single agent immune checkpoint inhibitor) may offer a way to prime the immune system hence obtaining a

synergistic response without encountering severe toxicities. This approach merits further clinical investigation.

The treatment paradigm and anticipated survival for patients with R/M HNSCC is rapidly changing. KEYNOTE-048 demonstrated a median overall survival of 12.3 months for patients with head and neck cancer treated with first line immunotherapy. However, this study included exclusively newly diagnosed platinum sensitive disease⁴. A more appropriate contemporary comparator population for this study is the CheckMate-141 trial evaluating nivolumab in platinum refractory R/M HNSCC of which 55% of patients had greater than one previous line of systemic therapy. In this trial, the median overall survival in patients treated with nivolumab was 7.5 months versus 5.1 months in patients treated with standard of care chemotherapy²⁴. Our study demonstrates a median overall survival of 9.8 months in a heavily pretreated population of which 61% received greater than one line of systemic therapy, 61% of patients were platinum refractory, and 42% of patients were refractory or PD-1 inhibitors.

This result is surprising due to the complex array of genetic alterations observed in advanced HNSCC patients. For example, through the available genomic data in this study, we identified two patients with tumors containing *KDR* (*VEGFR2*) mutations. Unfortunately, the functional significance of these alterations are currently unknown, even though this understanding would be important to help elucidate whether the positive effects of axitinib were due to function on tumor cells or to supporting cells in the microenvironment. For example, because one of these patients responded to therapy, if the *KDR* mutations are found to be activating and sufficient to make the protein resistant to axitinib, then the clinical data would suggest that inhibition of VEGF/VEGFR signaling in the tumor microenvironment may be more critical than inhibition of *KDR* signaling in tumor cells. As such, this trial opens an exciting area of research related to the pivotal for of VEGF/VEGFR signaling in HNSCC.

Importantly, we also the first to report a clinical link between PI3K status and response to axitinib. Given that approximately 45% of HNSCC harbor PI3K pathway alterations, future studies are warranted to evaluate whether PI3K pathway alterations are predictive of response to axitinib and potential mechanistic links between the two pathways in HNSCC. Multiple potential mechanisms may account for the relationship, for example, tumors with PI3K alterations are often induce angiogenesis through VEGFregulated cytokine mechanisms, and perhaps this process is critical for the survival of PI3K-dependent tumors²⁶. While future studies are necessary to help dissect the relationship between these two pathways, our discovery has the potential for profound clinical impact in this patient population and should be evaluated in larger patient cohorts. Although our study supports the activity of axitinib in heavily pretreated R/M HNSCC, there are limitations. The population was somewhat heterogeneous both in sites of primary disease and previous treatments. Patients with cutaneous squamous cell carcinoma (cSCC) are often excluded from R/M HNSCC given a distinct disease course and longer survival^{27, 28}. To evaluate this potential confounding factor, we evaluated the survival of the six patients with cSCC and found they had a worse OS, although not statistically significant, compared to the non-cSCC patients, in keeping with the low response rate within this subgroup. Hence, we do not believe this limits the interpretation of our results. Finally, given the improvement in OS with use of PD-1 inhibitors in heavily pretreated R/M HNSCC and the fact that only 11 patients (39.2%) were treated with a checkpoint inhibitor prior to enrollment, we questioned the role of potential receipt of a PD-1 inhibitor as a subsequent line of therapy as influencing survival within our study population. Eleven patients received a PD-1 inhibitor as some line of therapy after progression on axitinib. Exploratory analysis demonstrated no difference in survival between those subsequently treated with a checkpoint inhibitor versus those who were not suggesting this was not a confounding variable. Uniform inclusion criteria for previous treatments should be employed evaluating VEGF inhibition in futures studies.

Conclusion

Axitinib treatment is associated with improved survival in patients with heavily pre-treated head and neck cancer. The Choi Criteria were able to classify treatment responses amongst patients with an atypical radiographic response and should be considered for use in future trials of VEGFR directed tyrosine kinase inhibitors in head and neck cancer. Exploratory analysis suggests that marked response rates are seen with the use of a single agent ICI after axitinib (RR: 45%) and patients with PI3K pathway alterations may derive exceptional benefit from therapy (RR: 75% vs. 17%). Further investigation is warranted to evaluate its activity in biomarker selected populations, especially as a mechanism to prime the immune microenvironment prior to immune checkpoint inhibitor therapy.

References

1. Chaturvedi AK, Engels EA, Pfeiffer RM, et al. Human papillomavirus and rising oropharyngeal cancer incidence in the United States. J Clin Oncol. 2011;29: 4294-4301.

2. Siegel RL, Miller KD, Jemal A. Cancer statistics, 2020. CA Cancer J Clin. 2020;70: 7-30.

3. Mandal R, Senbabaoglu Y, Desrichard A, et al. The head and neck cancer immune landscape and its immunotherapeutic implications. JCI Insight. 2016;1: e89829.

4. Burtness B, Harrington KJ, Greil R, et al. Pembrolizumab alone or with chemotherapy versus cetuximab with chemotherapy for recurrent or metastatic squamous cell carcinoma of the head and neck (KEYNOTE-048): a randomised, open-label, phase 3 study. Lancet. 2019;394: 1915-1928.

5. Cancer Genome Atlas N. Comprehensive genomic characterization of head and neck squamous cell carcinomas. Nature. 2015;517: 576-582.

6. Strauss L, Volland D, Kunkel M, Reichert TE. Dual role of VEGF family members in the pathogenesis of head and neck cancer (HNSCC): possible link between angiogenesis and immune tolerance. Med Sci Monit. 2005;11: BR280-292.

7. Hasina R, Whipple ME, Martin LE, Kuo WP, Ohno-Machado L, Lingen MW. Angiogenic heterogeneity in head and neck squamous cell carcinoma: biological and therapeutic implications. Lab Invest. 2008;88: 342-353.

8. Smith BD, Smith GL, Carter D, Sasaki CT, Haffty BG. Prognostic significance of vascular endothelial growth factor protein levels in oral and oropharyngeal squamous cell carcinoma. J Clin Oncol. 2000;18: 2046-2052.

9. Kyzas PA, Cunha IW, Ioannidis JP. Prognostic significance of vascular endothelial growth factor immunohistochemical expression in head and neck squamous cell carcinoma: a meta-analysis. Clin Cancer Res. 2005;11: 1434-1440.

10. Ongkeko WM, Altuna X, Weisman RA, Wang-Rodriguez J. Expression of protein tyrosine kinases in head and neck squamous cell carcinomas. Am J Clin Pathol. 2005;124: 71-76.

11. Bran B, Bran G, Hormann K, Riedel F. The platelet-derived growth factor receptor as a target for vascular endothelial growth factor-mediated anti-angiogenetic therapy in head and neck cancer. Int J Oncol. 2009;34: 255-261.

12. Swiecicki PL, Zhao L, Belile E, et al. A phase II study evaluating axitinib in patients with unresectable, recurrent or metastatic head and neck cancer. Invest New Drugs. 2015;33: 1248-1256.

13. Choi H, Charnsangavej C, Faria SC, et al. Correlation of computed tomography and positron emission tomography in patients with metastatic gastrointestinal stromal tumor treated at a single institution with imatinib mesylate: proposal of new computed tomography response criteria. J Clin Oncol. 2007;25: 1753-1759.

14. Herbst RS, Arquette M, Shin DM, et al. Phase II multicenter study of the epidermal growth factor receptor antibody cetuximab and cisplatin for recurrent and refractory squamous cell carcinoma of the head and neck. J Clin Oncol. 2005;23: 5578-5587.

15. Motzer RJ, Escudier B, Tomczak P, et al. Axitinib versus sorafenib as second-line treatment for advanced renal cell carcinoma: overall survival analysis and updated results from a randomised phase 3 trial. Lancet Oncol. 2013;14: 552-562.

16. Hutson TE, Lesovoy V, Al-Shukri S, et al. Axitinib versus sorafenib as first-line therapy in patients with metastatic renal-cell carcinoma: a randomised open-label phase 3 trial. Lancet Oncol. 2013;14: 1287-1294.

17. Wolchok JD, Hoos A, O'Day S, et al. Guidelines for the evaluation of immune therapy activity in solid tumors: immune-related response criteria. Clin Cancer Res. 2009;15: 7412-7420.

18. Vermorken JB, Trigo J, Hitt R, et al. Open-label, uncontrolled, multicenter phase II study to evaluate the efficacy and toxicity of cetuximab as a single agent in patients with recurrent and/or metastatic squamous cell carcinoma of the head and neck who failed to respond to platinum-based therapy. J Clin Oncol. 2007;25: 2171-2177.

19. Vermorken JB, Mesia R, Rivera F, et al. Platinum-based chemotherapy plus cetuximab in head and neck cancer. N Engl J Med. 2008;359: 1116-1127.

20. Terme M, Pernot S, Marcheteau E, et al. VEGFA-VEGFR pathway blockade inhibits tumor-induced regulatory T-cell proliferation in colorectal cancer. Cancer Res. 2013;73: 539-549.

21. Doedens AL, Phan AT, Stradner MH, et al. Hypoxia-inducible factors enhance the effector responses of CD8(+) T cells to persistent antigen. Nat Immunol. 2013;14: 1173-1182.

22. de Almeida PE, Mak J, Hernandez G, et al. Anti-VEGF Treatment Enhances CD8(+) T-cell Antitumor Activity by Amplifying Hypoxia. Cancer Immunol Res. 2020.

23. Seiwert TY, Burtness B, Mehra R, et al. Safety and clinical activity of pembrolizumab for treatment of recurrent or metastatic squamous cell carcinoma of the head and neck (KEYNOTE-012): an open-label, multicentre, phase 1b trial. Lancet Oncol. 2016;17: 956-965.

24. Ferris RL, Blumenschein G, Jr., Fayette J, et al. Nivolumab for Recurrent Squamous-Cell Carcinoma of the Head and Neck. N Engl J Med. 2016;375: 1856-1867.

25. Taylor MH RD, Brose MS, et al. . A phase 1b/2 trial of lenvatinib plus pembrolizumab in patients with squamous cell carcinoma of the head and neck. J Clin Oncol. 2018;36: 6016.

26. Karar J, Maity A. PI3K/AKT/mTOR Pathway in Angiogenesis. Front Mol Neurosci. 2011;4: 51.

 Manyam BV, Garsa AA, Chin RI, et al. A multi-institutional comparison of outcomes of immunosuppressed and immunocompetent patients treated with surgery and radiation therapy for cutaneous squamous cell carcinoma of the head and neck. Cancer. 2017;123: 2054-2060.
 Venables ZC, Autier P, Nijsten T, et al. Nationwide Incidence of Metastatic Cutaneous Squamous Cell Carcinoma in England. JAMA Dermatol. 2019;155: 298-306.

Tables

Table 1: Patient Demographics and Clinical Characteristics

This table describes the baseline demographics of the patients included in analysis for

emcacy.		
Age	n	28
	Mean	63.9
	Median (range)	64.5 (37-80)
Gender, n (%)	Male	25 (89%)
Π	Female	3 (11%)
ECOG Performance Status, n (%)	0 (Fully functional)	11 (39%)
	1 (Minor Impairment)	17 (61%)
Disease Primary Site, n (%)	Oral Cavity	2 (7.1%)
~	Oropharynx	13 (46%)
	Larynx	4 (13.3%)
\mathbf{O}	Nasopharynx	3 (10.7%)
	Cutaneous	6 (21.4%)
HPV Status, n (%)	Positive	10 (35.7%)
	Negative	17 (60.7%)
	Unknown	1 (3.6%)
Previous Lines of Therapy	0	11 (39%)
	1	6 (21.4%)
	2	5 (17.8%)
	3+	6 (21.4%)
Previous Exposure to Platinum	Sensitive	11 (39.2%)
	Refractory	17 (60.7%)

Previous Exposure to PD-1 Inhibitor	n	11 (39.2%)
	Primary Resistant	3 (27%)
	Acquired Resistance	8 (54%)
	I	I



Table 2: Treatment Related Toxicities

This table demonstrates the toxicities observed in the entire study population (n=29) with a frequency of greater than 10%

Toxicity	Grade 1 or 2	Grade 3 or 4	All Grades
Fatigue	15 (54%)	6 (21%)	21 (75%)
Hypertension	13 (46%)	2 (7%)	15 (54%)
Oral Mucositis	2 (7%)	2 (7%)	4 (14%)
Diarrhea	6 (21%)	1 (4%)	7 (25%)
Oral pain	2 (7%)	1 (4%)	3 (11%)
Bleeding	4 (14%)	1 (4%)	5 (18%)
Nausea	9 (32%)	0 (0%)	9 (32%)
Weight loss	7 (25%)	0 (0%)	7 (25%)
Anorexia	6 (21%)	0 (0%)	6 (21%)
Aspartate aminotransferase increased	6 (21%)	0 (0%)	6 (21%)
Dysgeusia	5 (18%)	0 (0%)	5 (18%)
Vomiting	5 (18%)	0 (0%)	5 (18%)
Hoarseness	4 (14%)	0 (0%)	4 (14%)
Sore throat	4 (14%)	0 (0%)	4 (14%)
Dehydration	3 (11%)	0 (0%)	3 (11%)

Table 3: Treatment Efficacy

This table describes the efficacy and outcomes among A) evaluable patients and b) those with

sequencing results

Α			
6 month PFS (95% CI)	1	32% (18%,51%)	
Median PFS, days (95	% CI) ¹	107.5 (72-164)	
\bigcirc		3.5 months	
6 month OS (95% CI) ¹		71% (53%, 85%)	
Median OS, days KM	estimate (95% CI) ¹	301 (182,372)	
\mathbf{O}		9.8 months	
Best Overall Respons	se Rate	42%	
0)	Progressive Disease (PD), n (%)	10 (36%)	
	Stable Disease (SD), n (%)	3 (11%)	
	Partial Response (PR), n (%)	11 (39%)	
	Complete Response (CR), n(%)	1 (3%)	
	Off Treatment before 8 week scan, n (%)	3 (11%)	

1- 6 month survival proportion and 95% confidence interval estimated using Wilson score interval

method.

В			
		Response	e Rate (#
		responders	s/patients)
		Mutant	Wild-Type
PI3K Signaling Path	way Alterations	75% (6/8)	17% (2/12)
	Non-Cutaneous Squamous Cell Carcinoma	86% (6/7) ^A	12% (1/8)
	Cutaneous Squamous Cell Carcinoma	0% (0/1)	25% (1/4) ^в
KMT2C/D Mutation	ns	33% (2/6)	50% (6/12)
	Non-Cutaneous Squamous Cell Carcinoma	66% (2/3) ^c	50% (5/10)
	Cutaneous Squamous Cell Carcinoma	0% (0/3)	50% (1/2) ^в

^A Remaining patient had SD as best response to therapy.

^B Patient had a KDR (VEFGR2) S110F mutation and exhibited a complete response

^c Both patients who exhibited a response had synchronous mutations in the PI3K signaling pathway

Figure Legends

<u>Figure 1</u>: Kaplan Meier Survival Analysis- These figures illustrates the overall survival (1a) and progression free survival (1b) amongst patients treated with axitinib

<u>Figure 2</u>: Degree of Tumor Response- These figures demonstrate the maximal degree of response to treatment by Choi Criteria amongst evaluable patients (2a) as well as those with genomic sequencing results, clustered by mutation status (2b)

<u>Figure 3</u>: Genomic Alterations of Patient Cohort- This figure illustrates alteration status of selected genes of interest amongst evaluable patients with sequencing results.

Author Manu

Original Article

Axitinib in Head and Neck Cancer/Swiecicki et al

Efficacy of Axitinib in Metastatic Head and Neck Cancer With Novel Radiographic

Response Criteria<zaq;1>

Paul L. Swiecicki, MD¹⁻³<zaq;2>

Emily L. Bellile, MS⁴

Collin Brummel, BS⁵

J. Chad Brenner, PhD^{3,5}

Francis P. Worden, MD^{1,3}

¹Division of Hematology/Oncology, Department of Internal Medicine, University of Michigan Medical School, Ann Arbor, Michigan.

²Division of Hematology/Oncology, Department of Internal Medicine, Ann Arbor Veterans Affairs Medical Center, Ann Arbor, Michigan.

³Rogel Cancer Center, University of Michigan Medical School, Ann Arbor, Michigan. ⁴Department of Biostatistics, University of Michigan School of Public Health, Ann Arbor, Michigan.

⁵Department of Otolaryngology–Head and Neck Surgery, University of Michigan Medical School, University of Michigan Health System, Ann Arbor, Michigan. <FNTX>Additional supporting information may be found online in the Supporting

Information section at the end of the article.

We are grateful to the patients and families, without whom this study would not have been possible. We are particularly appreciative of the clinical and research staff assisting with the care of these patients.

Corresponding Author: Paul L. Swiecicki, MD, University of Michigan Medical School, 1500 E Medical Center Dr, C340, Ann Arbor, MI 48109-5848 (pswiecic@med.umich.edu).

Received July 3, 2020; revision received July 29, 2020; accepted August 20, 2020.</FNTX>

BACKGROUND: There are limited treatment options for unresectable recurrent or metastatic (R/M) head and neck squamous cell carcinoma (HNSCC). Vascular endothelial growth factor is of significant interest for targeted therapy in R/M HNSCC

because of its central role in tumorigenesis and immunosuppression. Axitinib is a potent inhibitor of vascular endothelial growth factor receptor 1 (VEGFR1), VEGFR2, VEGFR3, platelet-derived growth factor receptor<zaq;3>, and c-kit<zaq;4> and offers such an approach.

METHODS: This article reports the results of a phase 2 trial evaluating axitinib in R/M HNSCC according to the Choi criteria for radiographic response assessment.<zaq;5> The primary endpoint of this trial was 6-month overall survival.

RESULTS: Twenty-nine patients were enrolled, and 28 were evaluable for a response. Patients were heavily pretreated, with 61% having had at least 1 previous systemic treatment in the metastatic setting (range, 0-5). The median overall survival of 9.8 months and the 6-month overall survival rate of 70%<zaq;3> met the protocol-defined criteria for clinical efficacy. The best overall response rate was 42%. Correlative analyses demonstrated that PI3K signaling pathway alterations were associated with an increased response to therapy (75% vs 17%). A marked response to therapy was seen in a subgroup of patients who were treated with an immune checkpoint inhibitor after progression on axitinib.

CONCLUSIONS: Treatment with axitinib is associated with improved survival in patients with heavily pretreated head and neck cancer, and PI3K pathway alterations may serve as a biomarker for response. Further investigation is warranted to evaluate axitinib in biomarker-selected populations, especially in combination with immune checkpoint inhibitor therapy.

LAY SUMMARY:<zaq;6>

• Metastatic head and neck squamous cancer is an incurable disease with limited treatment options and a poor prognosis.

• This study is the first to demonstrate that the targeted oral drug axitinib improves survival in patients with heavily pretreated metastatic head and neck cancer.

• Furthermore, patients whose tumors have specific mutations derive the greatest benefit from therapy.

• The investigation of axitinib alone or in combination with immunotherapy in a genomic biomarker–selected population is warranted.

KEYWORDS: axitinib, Choi criteria, head and neck cancer, PI3K, vascular endothelial growth factor receptor inhibitor.

<H1>INTRODUCTION</H1>

Head and neck squamous cell carcinoma (HNSCC) is the sixth most common cancer with 600,000 new cases worldwide each year, and the incidence rate is increasing at an unprecedented rate because of the high prevalence of human papillomavirus–induced HNSCC.¹ In fact, oropharyngeal cancer is 1 of only 4 cancers increasing in incidence in the United States.² Although the majority of patients with HNSCC are cured with multimodality therapy, a significant proportion of patients develop unresectable recurrent or metastatic (R/M) HNSCC. Despite the recent development of programmed death 1 (PD-1) inhibitors, response rates remain low because of variability within the immune microenvironment.³ Even with these novel therapies, the median survival for patients newly diagnosed with R/M HNSCC is approximately 12 months.⁴

With increasing molecular characterization of HNSCC, there has been significant interest in targeted therapy.⁵ Vascular endothelial growth factor (VEGF) dysregulation has been identified as a crucial process in R/M HNSCC in not only angiogenesis but also progression, immunosuppression, and immune tolerance.^{6,7} Furthermore, VEGF overexpression is associated with advanced disease and a poor prognosis.^{8,9} Because of this central role in advanced disease and tumorigenesis, VEGF inhibition is of significant interest as a candidate for targeted therapy.

Axitinib is a multireceptor tyrosine kinase inhibitor<zaq;7> approved for renal cell carcinoma that inhibits several isoforms of the VEGF receptor (vascular endothelial growth factor receptor 1 [VEGFR1], VEGFR2, and VEGFR3). Furthermore, it has inhibitory activity against platelet-derived growth factor receptor (PDGFR)<zaq;3> and downstream effectors of epidermal growth factor receptor (EGFR), both of which are commonly disrupted and contribute to head and neck tumorigenesis.^{5,10,11} Because of this mechanism of action and known molecular alterations in R/M HNSCC, it seems to be a promising agent for clinical assessment.

We previously reported a phase 2 study evaluating axitinib in patients with heavily pretreated R/M HNSCC. This work demonstrated a low response rate (7%) with single-agent axitinib; however, a significant proportion of patients had stable disease (70%) with

radiographic findings consistent with a treatment response.¹² Moreover, the population had impressive overall survival (10.9 months), which suggested that efficacy was perhaps not captured. Hence, we postulated that axitinib held significant antitumor activity in R/M HNSCC, but the Response Evaluation Criteria in Solid Tumors (RECIST) failed to appropriately capture responders and may have inappropriately suggested tumor progression. Differential manifestations of response have been seen with the use of tyrosine kinase inhibitors (ie, swelling and cystic attenuation) that have the potential of being wrongly interpreted<zaq;8> as progressive disease by RECIST, and this prompted the development of the Choi criteria.¹³

On the basis of these findings, we initiated a new follow-up phase 2 study to investigate the clinical activity of axitinib in R/M HNSCC with the Choi criteria for response assessment. Our hypothesis was that axitinib would have significant antitumor activity as judged by the Choi criteria and would result in an improvement in the 6-month overall survival in comparison with a historical control.

<H1>MATERIALS AND METHODS</H1>

<H2>Patient Eligibility</H2>

This was a phase 2, open-label trial approved by the institutional review board of the University of Michigan Rogel Cancer Center (NCT02762513). All patients provided written informed consent. Patients 18 years old or older with histologically documented unresectable R/M HNSCC were eligible. All patients were required to have measurable disease according to a computed tomography scan or cutaneous lesions ≥ 10 mm that were not assessable on imaging but were present on physical examination, an Eastern Cooperative Oncology Group performance status of 0 to 2, and a life expectancy ≥ 12 weeks. Adequate hematopoietic, hepatic, and renal function was required, and this was defined as an absolute neutrophil count $\geq 1.5 \times 10^9$ cells/mL, a platelet count $\geq 75,000$ cells/mm³, a hemoglobin level ≥ 9.0 g/dL,<zaq;3> a total serum bilirubin concentration within 1.5 times the upper limit of normal, aspartate aminotransferase and alanine aminotransferase concentrations within 2.5 times the institutional upper limits of normal (unless there were liver metastases, in which case the aspartate aminotransferase and alanine aminotransferase concentrations had to be within 5.0 times the upper limit of normal), a serum creatinine clearance ≥ 30 mL/min, and a urinary protein level < 2+.

Women of childbearing potential must have had a negative serum or urine pregnancy test within the 3 days before treatment.

Patients who had tumors encasing major blood vessels or active hemoptysis (>0.5 teaspoons of bright red blood per day) or were currently using therapeutic anticoagulation were excluded, as were those with gastrointestinal abnormalities resulting in impaired absorption. Treatment with EGFR inhibitors within the 30 days preceding study entrance was prohibited. Patients were excluded if they had uncontrolled hypertension before enrollment, which was defined as a systolic blood pressure reading > 140 mm Hg and/or a diastolic blood pressure reading > 90 mm Hg.

<H2>Treatment Plan</H2>

Enrolled patients underwent a complete history and physical examination, baseline laboratory studies (complete blood count with differential, comprehensive metabolic profile, thyroid-stimulating hormone, and urinalysis),<zaq;9> and radiographic staging studies (neck/chest computed tomography and others as clinically warranted). If cutaneous lesions were not assessable for a response by imaging, pictures of the target lesions were obtained as well. All screening assessments were completed within the 28 days before the start of treatment.

Patients were initiated on axitinib at 5 mg twice daily with a cycle length of 28 days.<zaq;10> Dose escalation was planned at 2 weeks (to 7 mg twice daily) and 3 weeks for a goal of 10 mg twice daily in the absence of grade 2 or higher toxicities. Patients were seen for toxicity and laboratory assessments (complete blood count, comprehensive metabolic panel, thyroid-stimulating hormone, and urinalysis)<zaq;11> at 2 and 4 weeks and then monthly after treatment initiation. Response to therapy.<zaq;12> Dose escalation could be resumed at the next visit if toxicities diminished to grade 1 or lower. Treatment was continued until disease progression, unacceptable toxicity, or patient withdrawal of consent or at the discretion of the investigator.

<H2>Evaluation of Response</H2>

A response assessment was performed after 2 cycles of axitinib treatment, and this was continued every 2 cycles. Radiographic assessments obtained at enrollment were obtained at each time point. Similarly, if a physical examination was being used for the response assessment of cutaneous lesions, pictures were taken at each time point.

Photographs as well as imaging studies were submitted to the University of Michigan Tumor Response and Assessment Core. The radiologic response was determined according to the Choi criteria.¹³

<H2>Statistical Considerations</H2>

Twenty-nine patients were enrolled between August 30, 2016, and October 23, 2019. The median follow-up duration among the study participants was 18 months (range, 1-36 months), and no patients remained on therapy. Follow-up for patients still living ranged from 5 to 32 months. On the basis of our previous study supporting an improvement in survival for patients with R/M HNSCC treated with axitinib, we designed this expansion study. Although consideration was given to adjusting this original study to a Bayesian expansion trial design, it was ultimately decided<zaq;3> to begin a new cohort to test for an improvement in survival under the same mortality rate assumptions used in the previous study. Notably, treatment continuation decisions for this trial were based on Choi criteria that, as previously reported,¹² considerably differed from RECIST decisions when evaluated in the original trial.

The primary aim was to compare 6-month overall survival after treatment with axitinib in patients with unresectable R/M head and neck cancer with historical rates. On the basis of results in the literature, we assumed a 6-month mortality rate of 50% under current standard care in this patient population.¹⁴ A sample size of 37 patients was planned to test whether survival after treatment with axitinib was improved to 70% at 6 months in comparison with 50% with an upper tailed test of binomial proportion. No interim analyses for activity were planned. Because of an observed clinical benefit and a slowed accrual rate, an unplanned interim analysis was performed after the enrollment of 29 patients. Data were analyzed by the study statistician; a statistically meaningful improvement in survival was identified in this analysis, and the decision was made to close the study to further accrual.

Overall survival was defined as the time from study enrollment to death from any cause. Six-month overall survival was defined as the proportion of patients who received at least 1 cycle of axitinib<zaq;3> and were alive 6 months after study enrollment, and 95% confidence intervals (CIs) were estimated with the Wilson score interval method. Treatment-related adverse events were graded according to the Common Terminology

Criteria for Adverse Events, version 4.03. The response rate was defined as the sum of patients with complete responses and partial responses according to the Choi criteria. Statistical analysis was performed with SAS v14.3 software (SAS, Carey, North Carolina). Planned correlative analyses included a genomic analysis of patients when next-generation sequencing results were evaluable.

<H1>RESULTS</H1>

<H2>Patient Characteristics</H2>

Twenty-nine patients were enrolled, 1 of whom died before treatment with axitinib. All 28 patients who received at least 1 dose of axitinib were included for the toxicity analysis; the baseline characteristics are summarized in Table 1. The mean age was 63.9 years (range, 37-80 years), and the majority of the patients (61% [n = 17]) had an Eastern Cooperative Oncology Group performance score of 1, which indicated mild impairment. The primary site of disease for most patients was the oropharynx (42.6% [n = 13]),<zaq;13> and the majority of the study participants were negative for human papillomavirus (60.7% [n = 17]). The majority of the patients (61% [n = 17]) had at least 1 previous systemic treatment in the metastatic setting, with the number of previous lines of treatments ranging from 0 to 5. Seventeen patients (61%) were refractory to platinum therapy (defined as progression within 180 days of chemotherapy), and 12 patients (43%)<zaq;14> were previously treated with a PD-1 inhibitor.

<H2>Toxicity</H2>

The median duration of treatment was 3 cycles (range, 1-9 cycles). The most common toxicities included fatigue (75%), hypertension (54%), nausea (32%), and diarrhea (25%; Table 2). Bleeding was observed in 5 patients, including 1 patient with a grade 3 lower gastrointestinal bleed; all cases spontaneously resolved and recurred with the re-initiation of axitinib. Grade 3 or 4 severe toxicities were seen in 16 patients (57%). Severe toxicities included fatigue (21%), hypertension (7%), and mucositis (7%). No grade 5 events were reported. Overall, the observed toxicities were consistent with those previously reported in the literature.^{15,16}

<H2>Efficacy</H2>

The 6-month overall survival rate was 71% (95% CI, 53%-85%; Table 3). This met the protocol-defined criteria for supporting evidence of clinical benefit. The median

progression-free survival was 3.5 months (95% CI, 2.4-5.4 months), and the median overall survival was 9.8 months (95% CI, 5.9-12.2 months; Fig. 1).

Three patients completed their trial participation before response imaging: one on account of adverse effects (but the patient was clinically noted to have progressive disease),<zaq;3> another on account of death due to progressive disease, and a third on account of withdrawal from the study. The overall response rate was 42%,<zaq;15> and the disease control rate was 53%. The waterfall plot in Figure 2A graphically demonstrates the depth of response among participants evaluable for a response. One patient had a durable complete response. Only 1 patient with cutaneous squamous cell carcinoma demonstrated a response to therapy. This patient had a mutation in KDR (VEGFR2) and achieved a durable complete response. All of the remaining 6 patients with cutaneous squamous cell carcinoma had progressive disease.

Given the immunomodulatory potential of VEGFR inhibition, we evaluated the treatment response in patients who received PD-1 inhibition as part of their treatment course. Eleven patients were treated with a PD-1 inhibitor before treatment with axitinib. Three patients had primary resistance to checkpoint inhibitor therapy, and none of these patients responded to axitinib (0 of 3). Eight had acquired resistance to checkpoint inhibitor therapy: 3 of these patients had a partial response with axitinib (3 of 8 [37%]<zaq;16>), 2 had stable disease (2 of 8 [25%]), and 3 had progressive disease (3 of 8 [37%]). Eleven patients were treated with a PD-1 inhibitor after progression on axitinib with an observed response rate of 45% (5 of 11). The response assessment demonstrated a complete response in 1 patient (1 of 11 [9%]), a partial response in 4 patients (4 of 11 [36%]), and stable disease in 1 patient (1 of 11 [9%]); progressive disease was seen in the remaining 5 patients (5 of 11 [45%]).

<H2>Correlative Studies</H2>

To evaluate the association between genomic alterations, tumor characteristics, and clinical outcomes, we analyzed results from patients who had commercial next-generation sequencing previously performed (n = 20). The investigators defined a set of genes (sequenced as part of all next-generation sequencing panels), and recurrent alterations are shown (Fig. 3). Importantly, although no mutations were identified in FLT1 (VEGFR1), FLT4 (VEGFR3), PDGFR, or KIT, 2 patients had mutations<zaq;3> in

KDR (VEGFR2), including an S1100F mutation as well as 2 mutant alleles (R1032Q and G638R; Supporting Table 1). The ability of axitinib to inhibit these mutant forms of KDR is unknown; however, the patient with the S110F mutation had a complete response, whereas the other had progressive disease. Importantly, 55% of the patients (11 of 20) had TP53 alterations; 40% of the patients (8 of 20) harbored alterations to genes in the PI3K pathway, including PTEN and PIK3CA; and 30% of the patients (6 of 20) had mutations in either KMT2C (MLL2) or KMT2D (MLL3).

The degree of response and the pathway alterations were correlated for an exploratory analysis (Fig. 2B). The relative response rate<zaq;17> was 75% for patients with mutations in the PI3K pathway and 39% for wild-type patients (6 of 8 patients vs 2 of 12 patients). In terms of the KMT2C/D pathway, the response rate was 33% in the mutant population and 50% in the rest of the population (2 of 6 patients vs 6 of 12 patients). Because of the differential responses seen between patients with cutaneous squamous cell carcinoma and patients with noncutaneous primaries, the response rates were further explored (Table 3). Although sample sizes were limited, mutations in the PI3K pathway were associated with a higher response rate in comparison with the wild-type population in noncutaneous squamous cell carcinomas (86% vs 12%).

<H1>DISCUSSION</H1>

In this phase 2 study of patients with heavily pretreated unresectable R/M HNSCC, axitinib demonstrated an improvement in 6-month overall survival in comparison with historical controls (70% vs 50%). Furthermore, treatment resulted in significant response rates and lower rates of severe toxicities.

There is increasing recognition of variable radiographic manifestations of response with the advent of novel classes of therapeutics. Most recognized is the "pseudoprogression" observed with immunotherapy, which prompted the development of iRECIST to capture atypical responses.¹⁷ The Choi response criteria have been best evaluated in gastrointestinal stromal tumors, for which, in comparison with RECIST, they have been demonstrated to better predict survival.¹³ In our previous trial using RECIST, we observed a low RECIST-assessed objective response rate but paradoxically high and impressive overall survival among heavily pretreated patients.¹² As such, we hypothesized that we were underappreciating treatment responses with the use of

RECIST, and the Choi criteria may be more appropriate for discerning those patients benefiting from therapy. With the utilization of the Choi criteria in this study, we identified a response rate of 42%, with an additional 11% having stable disease. Furthermore, the use of these response criteria for treatment decisions resulted in an improvement in overall survival in comparison with historical controls, and this supports both that the Choi criteria appropriately identified treatment responders and that axitinib is an effective therapy for heavily pretreated patients with R/M HNSCC.

Targeted therapy has demonstrated promise in preclinical studies of HNSCC. Alterations in PI3KCA, CDKN2A, and EGFR suggest that head and neck cancer is a candidate for the development of targeted therapeutics. However, this approach has had limited clinical success. The only approved agent, cetuximab, has been demonstrated to improve survival by less than 3 months.^{18,19} Tyrosine kinase inhibitors offer the benefit of targeting numerous pathways (ie, VEGFR, EGFR, and PDGFR) and isoforms simultaneously. Axitinib has been demonstrated to inhibit VEGFR1, VEGFR2, and VEGFR3<zaq;18> as well as c-Kit. Mounting evidence suggests that VEGF inhibition is immunomodulatory via numerous mechanisms, including the production of interferon γ , reversal of the immunosuppressive microenvironment, and augmented activity of CD8+ T cells via hypoxia-inducible factor 1α secondary to tumor hypoxia.²⁰⁻²² Because VEGFR inhibition may prime the immune system for a response to immunotherapy, sequential use may be a modality to decrease toxicities yet still gain therapeutic synergy. In the small subgroup of patients who were treated with immunotherapy after axitinib (n = 11), the response rate<zaq;19> to PD-1 monotherapy was 45%; this included 1 patient with a complete response. Although conclusions cannot be drawn because of the limited sample size, previous trials have shown response rates<zaq;19> of 13% to 17% in biomarkerunselected populations,^{4,23,24} so this supports possible potentiation with sequential therapy. Preliminary results from the phase 1b/2 KEYNOTE-526 trial evaluating concurrent lenvatinib (an inhibitor of VEGFR1, VEGFR2, VEGFR3, FGFR1, FGFR2, FGFR3, FGFR4, and PDGFR α)<zaq;3> and pembrolizumab demonstrated a response rate of 40.9% and median progression-free survival of 8.2 months, which supported further investigation of this combination.²⁵

Fifty-seven percent of the patients in this study experienced grade 3 or 4 toxicities, the most common of which was fatigue. Large studies of the single-agent treatment regimens used in this patient population have demonstrated toxicity rates ranging from 35% to 46%.^{18,24} Although this study has a higher rate of serious toxicities in comparison with comparable agents, our previous study of single-agent axitinib demonstrated a much lower rate of severe toxicities (40%).¹² The toxicities encountered were manageable with dose reductions, and this supports patient tolerability. As previously mentioned, there is a promise of significant synergy with the combination of VEGF inhibition and PD-1 inhibition. Ongoing phase 3 trials are evaluating concurrent lenvatinib and pembrolizumab in patients with PD-L1 \geq 1%<zaq;20>. However, preliminary reports of clinical trials evaluating this combination describe grade 3 or 4 toxicities in 91% of patients, and this leads to 18% of study participants discontinuing treatment. Sequential therapy (ie, axitinib followed by a single-agent immune checkpoint inhibitor) may offer a way to prime the immune system and hence obtain a synergistic response without encountering severe toxicities. This approach merits further clinical investigation. The treatment paradigm and anticipated survival for patients with R/M HNSCC are rapidly changing. KEYNOTE-048 demonstrated a median overall survival of 12.3 months for patients with head and neck cancer treated with first-line immunotherapy. However, this study exclusively included newly diagnosed platinum-sensitive disease.⁴ A more appropriate contemporary comparator population for this study is the CheckMate-141 trial, which evaluated nivolumab in platinum-refractory R/M HNSCC; 55% of the patients had more than 1 previous line of systemic therapy. In this trial, the median overall survival was 7.5 months for patients treated with nivolumab and 5.1 months for patients treated with standard-of-care chemotherapy.²⁴ Our study demonstrated a median overall survival of 9.8 months in a heavily pretreated population in which 61% received more than 1 line of systemic therapy, 61% were refractory to platinum, and 42% were refractory to PD-1 inhibitors.<zaq;21>

This result is surprising because of the complex array of genetic alterations observed in patients with advanced HNSCC. For example, through the genomic data available in this study, we identified 2 patients with tumors containing KDR (VEGFR2) mutations. Unfortunately, the functional significance of these alterations is currently unknown, even

though this understanding would be important for elucidating whether the positive effects of axitinib are due to its function on tumor cells or supporting cells<zaq;22> in the microenvironment. For example, because 1 of these patients responded to therapy, if the KDR mutations are found to be activating and sufficient to make the protein resistant to axitinib, then the clinical data would suggest that inhibition of VEGF/VEGFR signaling in the tumor microenvironment may be more critical than inhibition of KDR signaling in tumor cells. As such, this trial opens an exciting area of research related to the pivotal for of VEGF/VEGFR signaling in HNSCC.

Importantly, we are also the first to report a clinical link between the PI3K status and the response to axitinib. Because approximately 45% of HNSCCs harbor PI3K pathway alterations, future studies are warranted to evaluate whether PI3K pathway alterations are predictive of a response to axitinib and potential mechanistic links between the 2 pathways in HNSCC. Multiple potential mechanisms may account for the relationship; for example, tumors with PI3K alterations often induce angiogenesis through VEGF-regulated cytokine mechanisms, and perhaps this process is critical for the survival of PI3K-dependent tumors.²⁶ Although future studies are necessary to help to dissect the relationship between these 2 pathways, our discovery has the potential for a profound clinical impact on this patient population and should be evaluated in larger patient cohorts.

Although our study supports the activity of axitinib in heavily pretreated R/M HNSCC, there are limitations. The population was somewhat heterogeneous in both sites of primary disease and previous treatments. Patients with cutaneous squamous cell carcinoma are often excluded from studies of R/M HNSCC<zaq;3> because of the distinct disease course and longer survival.^{27,28} To evaluate this potential confounding factor, we evaluated the survival of the 6 patients with cutaneous squamous cell carcinoma and found that they had worse overall survival, although this was not statistically significant, in comparison with the patients with noncutaneous squamous cell carcinoma<zaq;23>, and this was in keeping with the low response rate within this subgroup. Hence, we do not believe that this limits the interpretation of our results. Finally, given the improvement in overall survival with the use of PD-1 inhibitors in heavily pretreated R/M HNSCC and given the fact that only 11 patients (39.2%)<zaq;24>

were treated with a checkpoint inhibitor before enrollment, we questioned the role of the potential receipt of a PD-1 inhibitor as a subsequent line of therapy in influencing survival within our study population. Eleven patients received a PD-1 inhibitor as some line of therapy after progression on axitinib. An exploratory analysis demonstrated no difference in survival between those subsequently treated with a checkpoint inhibitor and those who were not, and this finding suggested that this was not a confounding variable. Uniform inclusion criteria for previous treatments should be used for evaluating VEGF inhibition in future studies.

In conclusion, axitinib treatment is associated with improved survival in patients with heavily pretreated head and neck cancer. The Choi criteria were able to classify treatment responses among patients with an atypical radiographic response and should be considered for use in future trials of VEGFR-directed tyrosine kinase inhibitors in head and neck cancer. An exploratory analysis suggests that marked response rates are seen with the use of a single-agent immune checkpoint inhibitor after axitinib (response rate<zaq;19>, 45%), and patients with PI3K pathway alterations may derive an exceptional benefit from therapy (response rate<zaq;19>, 75% vs 17%). Further investigation is warranted to evaluate its activity in biomarker-selected populations, especially as a mechanism for priming the immune microenvironment before immune checkpoint inhibitor therapy.

<H1>FUNDING SUPPORT</H1><zaq;25>

This study was approved and funded in part by the National Comprehensive Cancer Network Oncology Research Program from general research support provided by Pfizer, Inc.

<H1>CONFLICT OF INTEREST DISCLOSURES</H1>

Paul L. Swiecicki and J. Chad Brenner have applied for provisional patents (not awarded yet) for companion diagnostics related to the data presented in this publication. Swiecicki also reports grants from Ascentage Pharma and personal fees from Regeneron outside the submitted work. Francis P. Worden reports personal fees from Merck, Bristol-Myers Squibb, and Regeneron outside the submitted work. The other authors made no disclosures.

<H1>AUTHOR CONTRIBUTIONS</H1>

Paul L. Swiecicki: Conceptualization, data curation, formal analysis, methodology, resources, manuscript writing, and manuscript editing. Emily L. Bellile: Data curation, formal analysis, methodology, manuscript writing, and manuscript editing. Collin
Brummel: Data curation, formal analysis, and manuscript editing. J. Chad Brenner: Formal analysis, methodology, manuscript writing, and manuscript editing. Francis P. Worden: Conceptualization, resources, and manuscript editing.

REFERENCES

 Chaturvedi AK, Engels EA, Pfeiffer RM, et al. Human papillomavirus and rising oropharyngeal cancer incidence in the United States. J Clin Oncol. 2011;29:4294-4301.
 Siegel RL, Miller KD, Jemal A. Cancer statistics, 2020. CA Cancer J Clin. 2020;70:7-30.

3. Mandal R, Senbabaoglu Y, Desrichard A, et al. The head and neck cancer immune landscape and its immunotherapeutic implications. JCI Insight. 2016;1:e89829.

4. Burtness B, Harrington KJ, Greil R, et al. Pembrolizumab alone or with chemotherapy versus cetuximab with chemotherapy for recurrent or metastatic squamous cell carcinoma of the head and neck (KEYNOTE-048): a randomised, open-label, phase 3 study. Lancet. 2019;394:1915-1928.

5. Cancer Genome Atlas Network. Comprehensive genomic characterization of head and neck squamous cell carcinomas. Nature. 2015;517:576-582.

6. Strauss L, Volland D, Kunkel M, Reichert TE. Dual role of VEGF family members in the pathogenesis of head and neck cancer (HNSCC): possible link between angiogenesis and immune tolerance. Med Sci Monit. 2005;11:BR280-BR292.

7. Hasina R, Whipple ME, Martin LE, Kuo WP, Ohno-Machado L, Lingen MW. Angiogenic heterogeneity in head and neck squamous cell carcinoma: biological and therapeutic implications. Lab Invest. 2008;88:342-353.

8. Smith BD, Smith GL, Carter D, Sasaki CT, Haffty BG. Prognostic significance of vascular endothelial growth factor protein levels in oral and oropharyngeal squamous cell carcinoma. J Clin Oncol. 2000;18:2046-2052.

9. Kyzas PA, Cunha IW, Ioannidis JP. Prognostic significance of vascular endothelial growth factor immunohistochemical expression in head and neck squamous cell carcinoma: a meta-analysis. Clin Cancer Res. 2005;11:1434-1440.

10. Ongkeko WM, Altuna X, Weisman RA, Wang-Rodriguez J. Expression of protein tyrosine kinases in head and neck squamous cell carcinomas. Am J Clin Pathol. 2005;124:71-76.

11. Bran B, Bran G, Hormann K, Riedel F. The platelet-derived growth factor receptor as a target for vascular endothelial growth factor-mediated anti-angiogenetic therapy in head and neck cancer. Int J Oncol. 2009;34:255-261.

 Swiecicki PL, Zhao L, Belile E, et al. A phase II study evaluating axitinib in patients with unresectable, recurrent or metastatic head and neck cancer. Invest New Drugs. 2015;33:1248-1256.

13. Choi H, Charnsangavej C, Faria SC, et al. Correlation of computed tomography and positron emission tomography in patients with metastatic gastrointestinal stromal tumor treated at a single institution with imatinib mesylate: proposal of new computed tomography response criteria. J Clin Oncol. 2007;25:1753-1759.

14. Herbst RS, Arquette M, Shin DM, et al. Phase II multicenter study of the epidermal growth factor receptor antibody cetuximab and cisplatin for recurrent and refractory squamous cell carcinoma of the head and neck. J Clin Oncol. 2005;23:5578-5587.

15. Motzer RJ, Escudier B, Tomczak P, et al. Axitinib versus sorafenib as second-line treatment for advanced renal cell carcinoma: overall survival analysis and updated results from a randomised phase 3 trial. Lancet Oncol. 2013;14:552-562.

16. Hutson TE, Lesovoy V, Al-Shukri S, et al. Axitinib versus sorafenib as first-line therapy in patients with metastatic renal-cell carcinoma: a randomised open-label phase 3 trial. Lancet Oncol. 2013;14:1287-1294.

17. Wolchok JD, Hoos A, O'Day S, et al. Guidelines for the evaluation of immune therapy activity in solid tumors: immune-related response criteria. Clin Cancer Res. 2009;15:7412-7420.

18. Vermorken JB, Trigo J, Hitt R, et al. Open-label, uncontrolled, multicenter phase II study to evaluate the efficacy and toxicity of cetuximab as a single agent in patients with recurrent and/or metastatic squamous cell carcinoma of the head and neck who failed to respond to platinum-based therapy. J Clin Oncol. 2007;25:2171-2177.

19. Vermorken JB, Mesia R, Rivera F, et al. Platinum-based chemotherapy plus cetuximab in head and neck cancer. N Engl J Med. 2008;359:1116-1127.

20. Terme M, Pernot S, Marcheteau E, et al. VEGFA-VEGFR pathway blockade inhibits tumor-induced regulatory T-cell proliferation in colorectal cancer. Cancer Res. 2013;73:539-549.

21. Doedens AL, Phan AT, Stradner MH, et al. Hypoxia-inducible factors enhance the effector responses of CD8(+) T cells to persistent antigen. Nat Immunol. 2013;14:1173-1182.

22. de Almeida PE, Mak J, Hernandez G, et al. Anti-VEGF treatment enhances CD8(+)
T-cell antitumor activity by amplifying hypoxia. Cancer Immunol Res. 2020;8:806-818.
23. Seiwert TY, Burtness B, Mehra R, et al. Safety and clinical activity of
pembrolizumab for treatment of recurrent or metastatic squamous cell carcinoma of the
head and neck (KEYNOTE-012): an open-label, multicentre, phase 1b trial. Lancet
Oncol. 2016;17:956-965.

24. Ferris RL, Blumenschein G Jr, Fayette J, et al. Nivolumab for recurrent squamouscell carcinoma of the head and neck. N Engl J Med. 2016;375:1856-1867.

25. Taylor MH, Rasco DW, Brose MS, et al. A phase 1b/2 trial of lenvatinib plus pembrolizumab in patients with squamous cell carcinoma of the head and neck [abstract 6016]. J Clin Oncol. 2018;36:6016.

26. Karar J, Maity A. PI3K/AKT/mTOR pathway in angiogenesis. Front Mol Neurosci.2011;4:51.

27. Manyam BV, Garsa AA, Chin RI, et al. A multi-institutional comparison of outcomes of immunosuppressed and immunocompetent patients treated with surgery and radiation therapy for cutaneous squamous cell carcinoma of the head and neck. Cancer. 2017;123:2054-2060.

28. Venables ZC, Autier P, Nijsten T, et al. Nationwide incidence of metastatic cutaneous squamous cell carcinoma in England. JAMA Dermatol. 2019;155:298-306.

Figure 1. Kaplan-Meier survival analysis. This figure illustrates (A) overall survival and (B) progression-free survival among patients treated with axitinib.

Figure 2. Degree of tumor response. This figure demonstrates the maximal degree of response to treatment by the Choi criteria among (A) evaluable patients and (B) those with genomic sequencing results clustered by the mutation status. CR indicates complete response; PD, progressive disease; PR, partial response; SD, stable disease.<zaq;26>

Figure 3. Genomic alterations in the patient cohort. This figure illustrates the alteration status of selected genes of interest among evaluable patients with sequencing results.

Age (n = 28), y<zaq;27>  Mean 63.9  Median (range) 64.5 (37-80) Sex, No. (%)  Male 25 (89) 3(11)  Female ECOG performance status, No. (%)  0 (fully functional) 11 (39)  1 (minor impairment) 17 (61) Disease primary site, No. (%)  Oral cavity 2(7.1)  Oropharynx 13 (46)  Larynx 4 (13.3)  Nasopharynx 3 (10.7)  Cutaneous 6 (21.4) HPV status, No. (%)  Positive 10 (35.7)  Negative 17 (60.7)  Unknown 1 (3.6) Previous lines of therapy, No. (%)  0 11 (39)  1 6 (21.4)  2 5 (17.8)  ≥3 6 (21.4)

 TABLE 1. Patient Demographics and Clinical Characteristics

Abbreviations: ECOG, Eastern Cooperative Oncology Group; HPV, human

papillomavirus; PD-1, programmed death 1.

Previous exposure to PD-1 inhibitor (n = 11 [39.2%]), No. (%)

Previous exposure to platinum, No. (%)

 Sensitive

 Refractory

 Primary resistant

 Acquired resistance

This table describes the baseline demographics of the patients included in the analysis for efficacy.

11 (39.2)

17 (60.7)

3 (27) 8 (54)

TABLE 2.	Treatment-Related	Toxicities <zaq:2< th=""><th>8></th></zaq:2<>	8>

Toxicity	Grade 1 or 2, No. (%)	Grade 3 or 4, No. (%)	All Grades, No. (%)
Fatigue	15 (54)	6 (21)	21 (75)
Hypertension	13 (46)	2 (7)	15 (54)
Oral mucositis	2 (7)	2 (7)	4 (14)
Diarrhea	6 (21)	1 (4)	7 (25)
Oral pain	2 (7)	1 (4)	3 (11)
Bleeding	4 (14)	1 (4)	5 (18)
Nausea	9 (32)	0 (0)	9 (32)
Weight loss	7 (25)	0 (0)	7 (25)

Anorexia	6 (21)	0 (0)	6 (21)
Aspartate aminotransferase increased	6 (21)	0 (0)	6 (21)
Dysgeusia	5 (18)	0 (0)	5 (18)
Vomiting	5 (18)	0 (0)	5 (18)
Hoarseness	4 (14)	0 (0)	4 (14)
Sore throat	4 (14)	0 (0)	4 (14)
Dehydration	3 (11)	0 (0)	3 (11)

This table demonstrates the toxicities observed in the entire study population (n = 29)

with a frequency greater than 10%.

TABLE 3. Treatment Efficacy<zaq;29><zaq;30>

Evaluable Patients	Value		
6-mo PFS, % (95% CI) ^a	32 (18-51)		
PFS, median (95% CI), d ^a	107.5 (72-164)		
PFS, median, mo	3.5		
6-mo OS, % (95% CI) ^a	71 (53-85)		
OS (KM estimate), median (95% CI), d ^a	301 (182-372)		
OS, median, mo	9.8		
Best overall response rate, No. (%)	42% <zaq;31></zaq;31>		
Progressive disease	10 (36)		
Stable disease	3 (11)		
Partial response	11 (39)		
Complete response	1 (3)		
Off treatment before 8-wk scan	3 (11)		
	Response Rate, % (No. of Responders/No. of Patients)		
Patients With Sequencing Results	Mutant	Wild Type	
PI3K signaling pathway alterations	75 (6/8)	17 (2/12)	
Noncutaneous squamous cell carcinoma	86 (6/7) ^b	12 (1/8)	
:Cutaneous squamous cell carcinoma	0 (0/1)	25 (1/4)°	
KMT2C/D mutations	33 (2/6)	50 (6/12)	
Noncutaneous squamous cell carcinoma	66 (2/3) ^d	50 (5/10)	
Cutaneous squamous cell carcinoma	0 (0/3)	50 (1/2) ^c	

Abbreviations: CI, confidence interval; KM, Kaplan-Meier; OS, overall survival; PFS, progression-free survival.

This table describes the efficacy and outcomes among evaluable patients and patients with sequencing results.

^aThe proportions for 6-month survival and the 95% CIs were estimated with the Wilson score interval method.<zaq;32>

^bThe remaining patient had stable disease as the best response to therapy.

^cThe patient had a KDR (VEFGR2) S110F mutation and exhibited a complete response.

^dBoth patients who exhibited a response had synchronous mutations in the PI3K signaling pathway.

This study demonstrates that treatment with axitinib improves survival in patients with heavily pretreated recurrent or metastatic head and neck squamous cell carcinoma, that alternative response criteria enable the identification of patients with atypical radiographic responses, and that patients with PI3K pathway alterations may derive exceptional benefit from therapy. Clinically, this study provides evidence for the evaluation of axitinib alone or in combination with immunotherapy in a genomic biomarker–selected population.

AQ1: Please check the tables for accuracy in typesetting.

AQ2: Please confirm or correct the names and degrees of the authors, the affiliations, and the correspondence footnote. Please also provide a middle initial for Brummel if possible. AQ3: Please confirm or correct the sentence as edited.

AQ4: According to journal style, gene symbols (not the full names) should be italicized, whereas protein symbols should not be italicized. Please ensure that proper formatting is used for symbols throughout the article (no changes have been made).

AQ5: Please note that in accordance with journal style for abstracts, the first person ("we") has been eliminated from the abstract.

AQ6: Please confirm or correct the lay summary as edited; note that bullets have been added in accordance with journal style.

AQ7: Please confirm or correct the sentence as edited (especially the insertion of "inhibitor").

AQ8: Please confirm or correct "wrongly" (originally "abhorrently").

AQ9: Please confirm or correct the sentence as edited (especially "complete blood count" and "thyroid-stimulating hormone").

AQ10: Please verify the accuracy of the dosages.

AQ11: Please confirm or correct the sentence as edited (especially "complete blood count," "comprehensive metabolic panel," "thyroid-stimulating hormone," and "urinalysis").

AQ12: Please complete the unfinished sentence "Response to therapy" or delete it.

AQ13: Please confirm or correct "42.6% [n = 13]" (perhaps this should be "46% [n = 13]").

AQ14: Please confirm or correct "12 patients (43%)" (Table 1 cites 11 patients with respect to a PD-1 inhibitor).

AQ15: Here and elsewhere, please confirm or correct "42%" if this is based on 12 patients (12/28 = 0.429; ie, 43%); please also confirm or correct "53%" if this is based on 15 patients (15/28 = 0.5357; ie, 54%).

AQ16: Please confirm or correct "3 of 8 [37%]" (perhaps this should be "3 of 8 [38%]" or "3 of 8 [37.5%]").

AQ17: Please confirm or correct the sentence as edited (especially "wild-type patients"); please also confirm or correct "39%" (2/12 = 0.167; ie, 17%).

AQ18: Please confirm or correct "VEGFR1, VEGFR2, and VEGFR3" as edited as well as "c-Kit" ("c-kit" is cited in the abstract).

AQ19: Please confirm or correct "response rate" (originally "RR").

AQ20: Please confirm or correct "patients with PD-L1 \ge 1%" (perhaps this should be "patients with PD-L1 expression \ge 1%" or something else).

AQ21: Please confirm or correct the sentence as edited (especially "refractory to PD-1 inhibitors").

AQ22: Please confirm or correct "its function on tumor cells or supporting cells" as edited; please also rewrite "to the pivotal for of VEGF/VEGFR signaling" in the last sentence of this paragraph for greater clarity.

AQ23: Please confirm or correct "noncutaneous squamous cell carcinoma" (originally "non-cSCC").

AQ24: Please confirm or correct "39.2%" (perhaps this should be "39.3%" [11/28 = 0.3929]).

AQ25: Please confirm or correct the funding support section, the conflict-of-interest section, and the author contributions as edited.

AQ26: Please confirm or correct the figure legend as edited.

AQ27: In Table 1, please confirm or correct "Age (n = 28), y" and "Previous exposure to PD-1 inhibitor (n = 11 [39.2%])" as edited. Please also confirm or correct the data (for example, perhaps "n = 11 [39.2%]" should be "n = 11 [39.3%]" [11/28 = 0.3929]).

AQ28: According to the footnote, Table 2 concerns the entire study population of 29 patients, but the percentages seem to be based on 28 patients. If there is a discrepancy, please make any necessary changes.

AQ29: In Table 3, please confirm or correct the data (for example, perhaps "1 (3)" should be "1 (4)" [1/28 = 0.036]).

AQ30: In Table 3, please confirm or correct the addition of "Evaluable Patients" and "Patients With Sequencing Results" (as replacements for "A" and "B"). Please also confirm or correct the addition of "PFS, median, mo" and "OS, median, mo" and the editing of "Response Rate, % (No. of Responders/No. of Patients)."

AQ31: In Table 3, please consider replacing "42%" with an n value and "42" in parentheses.

AQ32: In Table 3, please confirm or correct the footnotes and their citations as edited.

Author Man

1A:Overall Survival











Author Ma

cncr_33226_f3.eps